DO PROTECTED AREAS REDUCE FOREST FRAGMENTATION? A MICROLANDSCAPES APPROACH

July, 2013

Katharine R. E. Sims

Department of Economics AC #2201, Amherst College Amherst, MA 01002-5000 Phone: (413) 542-2902 Fax: (413) 542-2090 ksims@amherst.edu

ABSTRACT

Conservation policies influence both the amount of habitat loss and patterns of habitat fragmentation. This paper develops a "microlandscapes" approach that combines fragmentation measures with quasi-experimental evaluation methods in order to assess the effects of policy on habitat fragmentation. As an application, the paper estimates whether and to what extent wildlife sanctuaries and national parks in Thailand prevented forest loss and fragmentation. I find that both types of protected areas significantly increased forest cover, average forest patch size and maximum forest patch size. Comparisons between the two types indicate that wildlife sanctuaries were more effective than national parks in terms of protecting forest in the interior vs. exterior areas of parks and preventing fragmentation conditional on the level of forest cover. The differences are consistent with predicted differences resulting from spatial patterns of enforcement that are uniform or core-focused in the wildlife sanctuaries versus boundary-focused or include agglomeration penalties in the national parks. Given the greater effectiveness of wildlife sanctuaries in preventing fragmentation and the suggestive link to enforcement types, these results reinforce existing theoretical work urging conservation managers to consider how the spatial distribution of enforcement may affect patterns of resource use.

JEL classifications: Q23, Q24, Q56, Q57

Keywords: habitat fragmentation, land use, land conservation, protected areas, policy evaluation

1. INTRODUCTION

The fragmentation of large contiguous areas of habitat into smaller, dispersed patches is an important driver of biodiversity losses worldwide (Turner 1996, Bender et al. 1998, Fahrig 2002, Fischer and Lindenmayer 2007). Habitat fragmentation also reduces the productivity of ecosystems because smaller areas of habitat are often less resilient to severe weather or disease shocks (Boose et al. 1994, Rochelle et al. 1999, Allan et al. 2003, Opdam and Wascher 2004). The design of efficient and effective land conservation policies must therefore consider both the total amounts of habitat conserved and the spatial configuration of that habitat (Albers 1996, Bockstael 1996, Albers and Bu 2009, Lewis and Alig 2009, Albers et al. 2010). Previous research considers the costs and benefits of different habitat patterns and simulates the expected impacts of future policies on habitat fragmentation (Lewis and Plantinga 2007, Horan et al. 2008, Lewis et al. 2009, Ando and Shah 2010, Lewis 2010, Lewis et al. 2011), but does not rigorously evaluate the *ex post* impacts of land conservation policies.

The purpose of this paper is to develop and apply an approach for retrospective empirical evaluation of policy impacts on habitat fragmentation. To do so, I divide regional landscapes into smaller units, or "microlandscapes," so that impacts can be assessed by comparing areas affected by the policy to otherwise similar areas. This approach is novel in its combination of methods for measuring habitat fragmentation (Heilman et al. 2002, Riitters et al. 2002, Stanfield et al. 2002, Butler et al. 2004, Alig et al. 2005, Hawbacker et al. 2005, 2006) with methods for quasi-experimental policy evaluation. Quasi-experimental methods have previously been used to evaluate the impacts of conservation policies on loss of habitat or species (Ferraro et al. 2007, Costello et al. 2008, Andam et al. 2008, 2010, Greenstone and Gayer 2009, Pfaff et al. 2009, Sims 2010), but they have not been used to evaluate the impacts of conservation policies on habitat fragmentation.

I apply the microlandscapes approach to assess whether and to what extent protected areas in Thailand prevented forest loss and fragmentation. Protected areas are one of the most common land conservation policies worldwide (Chape et al. 2005) and have the potential to significantly reduce habitat fragmentation by conserving large, contiguous areas of land. Yet the

majority of new protected areas are in developing countries (Zimmerer et al. 2004), where there is substantial pressure on natural resources and where legal restrictions are often incompletely enforced. To illustrate how spatial patterns of enforcement influence fragmentation in the context of protected forest areas, I use a simple von Thünen style model of clearing behavior on a sample microlandscape which is under pressure for conversion from forest to agricultural land. This framework illustrates that enforcement which is uniformly distributed in space or is focused on core areas of parks is likely to be more effective than enforcement which is stronger near park boundaries or includes agglomeration penalties, in terms of protecting forests in the interior areas of parks and preventing forest fragmentation conditional on the level of forest cover.

I then evaluate the impacts of strictly protected areas—wildlife sanctuaries and national parks-in North and Northeast Thailand. Wildlife sanctuaries are the most restrictive, aimed at preserving biodiversity and legally permitting only research or small-scale ecotourism activities. National parks are less restrictive, allowing for recreational use and tourism (Dixon and Sherman 1990, ICEM 2003, Chettamart 2003, FAO 2009). Both types of protected areas officially prohibit resource extraction and agricultural use, but in reality have been incompletely enforced and contain significant amounts of agricultural land (Vandergeest 1996, Roth 2004a,b, ICEM 2003, Chettamart 2003, FAO 2009).¹ Direct measures of enforcement are not available, but anecdotal evidence suggests that wildlife sanctuaries received more consistent funding and a clearer mandate to protect core habitat areas while national parks had fewer resources available for enforcement and were more lenient about subsistence use (Dixon and Sherman 1990, Albers and Grinspoon 1997, ICEM 2003, Chettamart 2003, Vandergeest 2003). This history raises the possibility of different enforcement levels and patterns between the two types of protected areas. The paper therefore examines both the overall effectiveness of wildlife sanctuaries and national parks in preventing forest loss and fragmentation as well as their differential effects on land-use patterns.

To estimate impacts, I use new panel data on protected area designations and forest cover between 1973 and 2000. I first divide regional landscapes into 5 km x 5 km microlandscapes and

¹ A 2003 report by the International Center for Environmental Management (ICEM) estimated that more than 500,000 people were living inside the strictly protected areas in Thailand. Legally, wildlife sanctuaries are similar to IUCN Category I areas while national parks are similar to IUCN Category II areas. The laws governing the two major types of protected areas and establishing the process for their designation were passed in the 1960's and remained in force without major changes throughout the period of this study.

use covariate matching to select control areas which were similar with respect to the characteristics that determined protected area locations and that might also influence fragmentation outcomes. I then estimate panel regressions including microlandscape fixed-effects, thereby identifying impacts using plausibly exogenous variation in the timing of protected area designations.

I evaluate the following outcomes for each microlandscape: the total amount of forest cover, average forest patch size, maximum forest patch size, cleared patch size, forest patch perimeter to area ratio, and the density of cleared patches. I select this set of fragmentation measures because they are believed to be ecologically relevant (Betts 2000, Leitao and Ahern 2002) and share a common framework based on habitat patches (Urban et al. 1987). Although there is debate about exactly which patterns of habitat maximize biodiversity or ecosystem productivity (Christensen 1997, Galvin et al. 2008), I define landscapes in this context as less fragmented if they have more forest cover and larger average and maximum forest patch sizes. In addition, I define them as less fragmented if they have larger cleared patches, conditional on the level of forest cover.²

I find that both wildlife sanctuaries and national parks in Thailand have prevented significant forest loss and fragmentation, increasing forest cover by an estimated 19%, average forest patch size by 25%, and maximum forest patch size by 21%, compared to the counterfactual of no protection. However, comparisons between the two types indicate that wildlife sanctuaries were more effective at preventing forest fragmentation. Wildlife sanctuaries appear to have retained greater forest cover and forest patch size in the interior areas than national parks. They also resulted in marginally significantly larger forest patch sizes,

² The literature on habitat fragmentation contains numerous possible metrics for assessing outcomes, due to the many possible configurations of habitat (O'Neill et al. 1988, Turner 1990, Gustafson 1998, McGarigal and Cushman 2002). The chosen metrics measure both size and shape of habitat patches (Betts 2000, McGarigal and Cushman 2002). Larger forest patch sizes are thought to be important for species which cannot easily cross deforested areas to forage or reproduce. Habitat patch shape is important for species which require a safe distance from the edge of a patch and thus prefer patches with more core habitat. Smaller forest patch perimeter to area ratios indicate more core habitat relative to edge habitat (picture the difference between a circle and an amoeba shape). Larger cleared patch sizes and a lower density of cleared patches also indicate more core forest habitat, i.e. forested landscapes which are less broken-up by many small, dispersed areas of clearing (picture a solid block of forest versus one with lots of holes). Forest patch size tends to increase with forest cover, but forest patch perimeter to area ratio and the density of cleared patches do not necessarily vary monotonically with forest cover (see Section 3.1 for an example), so are compared here conditional on forest cover.

significantly larger cleared patch sizes, significantly smaller forest patch perimeter to area ratios and significantly lower densities of cleared patches, conditional on the level of forest cover. Such differences are consistent with predicted differences between spatial enforcement patterns that are uniform or core-focused in the wildlife sanctuaries versus boundary-focused or include agglomeration penalties in the national parks. Given the greater effectiveness of wildlife sanctuaries in reducing forest fragmentation and the suggestive link to different enforcement types, these results reinforce existing theoretical work urging protected area managers to consider how the spatial distribution of enforcement may affect patterns of resource use (Albers and Grinspooon 1997, Robinson et al. 2002, 2005, 2008, Albers and Muller 2004, Albers 2010, Robinson and Lokina 2011).

The paper proceeds as follows. Section 2 briefly reviews previous literature and outlines the contributions of this paper in relation to that research. Section 3 describes the conceptual framework of clearing at the parcel level and applies this to a sample microlandscape to illustrate how different patterns of enforcement affect forest fragmentation. Section 4 describes the empirical strategy, including the data, the matching approach and the panel regression specification. Section 5 presents and discusses results and Section 6 concludes.

2. PREVIOUS LITERATURE

Conservation policies influence both the amount of habitat loss and patterns of habitat fragmentation. One strand of relevant previous literature explores the determinants of fragmentation and develops methods to predict and value policy outcomes. Another considers the effects of protected areas, including both theoretical and empirical assessments of their effectiveness. The microlandscapes approach brings these strands together by combining fragmentation metrics with policy evaluation techniques to analyze protected area impacts.

2.1 Empirical analysis of policies affecting fragmentation

Previous studies of the determinants of fragmentation suggest that land-owner type, zoning regulations and road building are important potential policy levers. In the United States, Butler et al. (2004) test whether forest fragmentation in western Oregon and Washington is linked to population density, income, agricultural profitability, distance to roads, slope, and federal land ownership. Alig et al. (2005) extend this approach by testing whether the spatial

configuration of returns within each landscape (as measured by the pattern of low-quality land) is also a driver of fragmentation. Both find that more federal land ownership reduced fragmentation while proximity to highways increased it.³ Turner et al. (1996) similarly shows a correlation between land ownership categories and land-cover change patterns in North Carolina and Washington. Albers, Ando and Chen (2008) find that federal ownership of conservation lands created crowding-in and crowding-out effects of private ownership (depending on the state), thereby also indirectly influencing fragmentation. The potential importance of zoning regulations for fragmentation patterns is suggested by the finding (Alig et al. 2005 and Butler et al. 2004) that fragmentation is positively related to population density and proximity to urban areas. Hawbacker et al. (2005, 2006) also find significant correlations between habitat patch size and housing density as well as biophysical characteristics and land use for Northern Wisconsin townships.

Several prior studies suggest the importance of roads as a driver of fragmentation. Heilman et al. (2002) is one of the first to develop fragmentation statistics for the U.S. based on road patterns which increase fragmentation over time. Albers et al. (2012) proposes a new "Road Network Agglomeration Index," which is a strong indicator of fragmentation in California's reserve network. Analyses of deforestation patterns in developing countries (Chomitz and Gray 1996, Pfaff 1999, Pfaff et al. 2007, 2011, Robalino and Pfaff 2012) also highlight the importance of roads. Although these papers do not measure forest fragmentation directly, they show that road density is a key driver of deforestation. Roads influence deforestation both directly, through decreased transportation costs and increased returns to deforestation (Chomitz and Gray 1996), and indirectly, through spatial or inter-temporal spillovers (Pfaff et al. 2007, Robalino and Pfaff 2012).

In addition to exploring the broad determinants of fragmentation, previous studies develop methods to predict how specific policies may affect fragmentation. A group of new studies (Lewis and Plantinga 2007, Lewis, Plantinga and Wu 2009, Lewis 2010, Lewis et al. 2011) forecasts ecosystem change using past econometric relationships between land uses and returns plus simulations of how policies change those returns. Lewis and Plantinga (2007) develop this method to analyze how afforestation subsidies with and without agglomeration

³ An earlier study of Oregon watersheds also finds that land ownership structure is related to size of forest patches (Stanfield et al. 2002).

bonuses affect fragmentation in South Carolina's coastal plain. Lewis, Plantinga and Wu (2009) extend the method and analyze a situation in which afforestation subsidies can also vary between areas with different initial landscape conditions. Lewis (2010) predicts the probability of a local species going extinct depending on different housing densities on Northern Wisconsin lakes. Finally, Lewis et al. (2011) uses a similar methodology to assess the efficiency of different types of incentive-based policies depending on the degree of returns to scale in large habitat parcels.

These studies provide important tools for making forward-looking assessments of policies which are likely to potentially affect habitat fragmentation. However, as noted by Lewis and Alig (2009) in their review of methods for modeling landscape change, there is also a need for work which uses quasi-experimental methods to retrospectively evaluate how specific policies have influenced fragmentation. Actual responses to policy may differ from predictions because people and economic systems respond in unforeseen ways. The microlandscapes approach described here attempts to fill this methodological gap in the literature by combining fragmentation metrics with empirical evaluation methods.

2.2 The impacts of protected areas on forest fragmentation

This paper also seeks to fill a gap in our specific knowledge of the effects of protected areas on fragmentation patterns. Previous related literature includes theoretical modeling of extraction and enforcement patterns, empirical analyses of avoided deforestation benefits, and explorations of the association between protection and fragmentation.

Several theoretical models in the literature illustrate the idea that resource users and protected area managers are involved in a complex spatial and temporal game (Albers and Grinspooon 1997, Robinson et al. 2002, 2005, 2008, Albers and Muller 2004, Albers 2010, Robinson and Lokina 2011). In general, these models analyze situations where resource users illegally enter protected forests to extract firewood or non-timber forest products. Robinson et al. (2002) show how changes in market access, minimum consumption needs, opportunity costs of labor and extraction costs affect the spatial patterns of resource extraction. They illustrate that policies which increase economic returns in extraction zones near protected areas may be more effective than increased enforcement because extraction decisions are very sensitive to labor costs and market access. Robinson et al. (2008) and Robinson and Lokina (2011) add a temporal dimension to the resource extraction decision and incorporate the possibility that communities

may cooperate to manage resources. A key conclusion with respect to fragmentation is that snapshots of resource patterns may not accurately capture dynamic community decisions about regeneration or changing responses over time to protected areas. Albers and Grinspoon (1997) and Albers and Muller (2004) consider the success of different types of management regimes in protecting forests. Albers and Muller (2004) show that the effectiveness of enforcement strategies including direct punishment, conservation payments or integrated development and conservation projects depends on the context in which they are applied. Finally, Robinson et al. (2011), Robinson and Lokina (2011) and Albers (2010) bring together resource users and managers in a spatial game-theoretic model. Albers (2010) considers best responses by each side given constraints about different spatial patterns of enforcement including homogeneous patrols, boundary patrols, interior ring patrols and enforcement zones. She illustrates that managers can more efficiently protect core areas, thereby lessening fragmentation of the interior, if they are able to spatially target enforcement and allow extraction zones at the edge of protected areas. Unfortunately, resource managers in many countries (including Thailand) are legally prohibited from allowing extraction zones or explicit spatial targeting, possibly locking-in inefficient management regimes.

Within the empirical literature on protected areas, most research has focused on whether protected areas are effective in limiting the total amount of forest loss. These studies include evaluations of deforestation in Thailand (Cropper et al. 1999, 2001, Sims 2010), Costa Rica (Andam et al. 2008, Pfaff et al. 2009, Ferraro et al. 2011), Mexico (Deininger and Minten 2002), Belize (Chomitz and Gray 1996) and Brazil (Pfaff 1999). Together, they do indicate that protected areas have prevented significant amounts of deforestation, but less than would be implied by naïve inside-outside comparisons. In addition, these and other studies (Joppa and Pfaff 2009) document that protected areas were usually designated in more remote and rugged locations where the threat of deforestation was lower. Failing to account for this selection issue leads to overestimates of park effectiveness in maintaining forest cover.

To date, studies which have measured fragmentation inside and outside of protected areas do not account for this location bias. Papers by Sanchez et al. (2003, 2001, 1999) develop remote sensing methods for measuring tropical deforestation over time. They include calculations for forest fragmentation inside and outside of protected areas in Costa Rica, but do not attempt to derive causal impact estimates of protected area policy. Liu et al. (2001) do imply that changes in

habitat fragmentation inside and outside of a panda reserve in China can be attributed to changes in protection status, but fail to account for other underlying differences. Several additional studies also measure forest fragmentation over time and draw possible links to community or state management regimes on the ground, but do not specifically examine protected areas. Examples of developing country analyses include Ochoa-Gaona (2001), Fox et al. (1995), Fox and Vogler (2005), and Thomas et al. (2004). In summary, previous empirical studies on the deforestation impacts of protected areas either do not measure fragmentation or compare fragmentation inside and outside of protected areas without accounting for other potentially confounding differences in land characteristics. This paper seeks to fill this gap by matching microlandscapes with similar characteristics and comparing changes over time using panel data analysis in order to rigorously analyze protected area impacts on fragmentation.

3. CONCEPTUAL FRAMEWORK

3.1 Clearing and fragmentation patterns on a sample landscape

Protected forest areas have the potential to reduce fragmentation by setting aside large contiguous areas of forest under public protection. Yet where those forests are under pressure for conversion to other uses and enforcement is not complete, deforestation and fragmentation will occur despite official restrictions on resource use. To illustrate the potential impacts of different incomplete enforcement patterns on forest loss and fragmentation, I construct a simple economic framework of clearing behavior on a sample two-dimensional microlandscape (illustrated in Figure 1). The microlandscape consists of multiple individual parcels; in this case 30 x 30 parcels. I follow the standard von Thünen style assumption that land-use choices on each parcel are driven by the relative returns to different land uses (e.g. Stavins and Jaffe 1990, Chomitz and Gray 1996, Pfaff 1999, Angelsen 2007, 2010). For simplicity, I assume that a parcel is cleared if the rents to agricultural use are weakly positive and otherwise remains forested. Agricultural rents are a function of the price of the agricultural product, p^a , times the yield to agriculture on each plot as a function of land quality, $f(q_i)$, minus labor costs (wage, w, times the quantity of labor, *l*), transportation costs as a function of distance to market $t(d_i)$ and clearing costs (*c*):

(1) $R_i^a = p^a f(q_i) - wl - t(d_i) - c$

Parcels within the microlandscape vary with respect to underlying land quality (q_i) and distance to market (d_i) as shown in panels a and b of Figure 1.⁴ Wages are assumed to be exogenous and the amount of labor needed for each parcel is fixed.⁵

Figure 1c and d show examples of clearing patterns produced by this framework (specific parameter values are given below Figure 1).⁶ The baseline scenario (Figure 1c) has a relatively low agricultural price and thus total amount cleared (10%). The second scenario (Figure 1d) has a higher agricultural price, modeling a rise in the value of the agricultural good over time which drives increased clearing (total amount cleared = 45%). A comparison of the patterns in Figures 1c and 1d illustrates changes over time consistent with the key stylized facts of tropical deforestation (e.g. Angelsen 2010) and similar to those observed in the actual Thai land-cover data. In general, deforestation increases where rents are higher, i.e. on higher quality land or land closer to markets. Forest fragmentation accompanies deforestation and is driven by the underlying patchy or branching pattern of land quality. Considering the specific set of fragmentation metrics which will be used in the empirical section, we note that as total clearing increases, forest patch size will generally decrease as the forest is divided into more, smaller patches. However, forest patch perimeter to area ratios and cleared patch densities may either increase or decrease with more clearing, depending on the underlying patchiness of land quality and the role of transportation costs.⁷

⁴ For simplicity, land quality is assumed to be a function of distance to the nearest water source, but additional sources of heterogeneity in land quality such as soil type, slope, aspect, distance to local villages or paths, etc. could also be important. Adding more dimensions to the land quality function would change the specific patterns on the microlandscape but not the overall conclusion that fragmentation will be driven by patchy or branching distributions of land quality.

⁵ This corresponds to an assumption that the microlandscape is small compared to the overall labor market and workers are freely mobile. Since this mobility assumption may not hold, the implications of relaxing it are also discussed in Section 3.3.

⁶ Parameter values are illustrative and are not based on calibrations. The predictions described below are based on tendencies from experimenting with a variety of different parameter values. Excel model available on request.

⁷ As an example, consider additional scenarios on the same landscape pictured in Figure 1. At p^a =.7, there would be is 1% cleared and 1 cleared patch; at p^a =.79, 5% cleared and 3 cleared patches; at p^a =.85, 10% cleared and 4 cleared patches; at p^a =1.01, 30% cleared and 5 cleared patches; at p^a =1.15, 50% cleared, 2 cleared patches and at p^a =1.36, 70% cleared and 1 cleared patch. This non-monotonic relationship happens because clearing initially spreads to more small dispersed areas and the density of cleared patches increases, but at high levels of deforestation the cleared patches merge together and cleared patch density decreases.

3.2 Protected areas and possible enforcement patterns

I now consider how clearing and fragmentation outcomes will vary as a result of protection with different spatial patterns of enforcement (Figure 2). Although protected area managers may be legally expected to fully enforce regulations, in reality enforcement is likely to be both incomplete and non-uniformly distributed across space (e.g. Albers 2010). I consider three possible patterns of enforcement: spatially uniform, boundary-focused and core-focused. In addition, I consider what happens if an agglomeration penalty is added to these patterns.

Each protection scenario (Figure 2) uses the same framework and sample microlandscape as above. The microlandscape is now assumed to be set aside as part of a protected area after the baseline clearing scenario (Figure 1c) but before the rise in agricultural prices. The entire microlandscape is within the protected area, with the park boundary at the bottom of the microlandscape (dotted black line). Parcels close to the top of the microlandscape are thus more in the interior of the park while those at the bottom are nearer to the boundary. Enforcement is assumed to be incomplete, so new clearing will accompany the rise in agricultural prices. In order to consistently compare fragmentation patterns across scenarios, the amount of enforcement is assumed to be such that it results in the same total amount of deforestation across the cases.

Uniform enforcement

Spatially uniform enforcement is the baseline enforcement type. To model this case, I assume that parcel rents depend additionally on the chance of being caught and penalized for clearing, where π_u is the probability of detection and G_u is the fine that must be paid if caught: (2) $E(R_i^a) = p^a f(q_i) - wl - t(d_i) - c - \pi_u G_u$

Since enforcement is uniformly applied, the probability of detection (π_u) is the same on every plot and thus rents are reduced evenly across space. Larger penalties (G_u) will result in less clearing. Figure 2a shows an example where spatially uniform enforcement results in 20% clearing.

Boundary-focused enforcement

A second possible enforcement pattern is boundary-focused enforcement. More frequent patrols near boundaries are likely where resources are limited and transportation is costly. Fujita (2003) reports clear resource limitations on patrols in Thailand: in 1989 there were about 240

forest protection units with 5-6 people per unit who were responsible for enforcing Royal Forestry Department policy for all of Thailand (some 517 thousand sq km). Road networks are likely to be better near the boundaries of parks and a reasonable manager might also expect more illegal activities near the boundaries. To model boundary-focused enforcement I assume that the probability of being caught and fined (π_b) varies with the distance to the boundary (b_i) of the protected area:

(3)
$$E(R_i^a) = p^a f(q_i) - wl - t(d_i) - c - \pi_b(b_i)G_b$$

where π_b is decreasing as distance to the boundary increases. An example, again with 20% clearing, is shown in Figure 2b.

Core-focused enforcement

A third enforcement scenario (Figure 2c) is core-focused enforcement. This situation might occur if the regulator is intentionally trying to protect habitat at the center of protected areas and so levies greater punishment in the interior. In Thailand, the wildlife sanctuaries were created explicitly for habitat conservation and preservation of game (Dixon and Sherman 1990, Chettamart 2003); managers may therefore have focused on keeping core areas intact in those areas. In addition, a key goal of Thai protected areas was to safeguard national water resources and prevent downstream flooding (Ruhle 1964, Hirsch and Lohmann 1989, Vandergeest 2003, ICEM 2003). Enforcement resources may also have been differentially focused on sensitive upper watershed areas which are more likely to be in the interiors of parks. I model core-focused enforcement with the same Equation (3) above, except that the probability of enforcement (denoted π_c) is now increasing as distance to the boundary increases, and the penalty is denoted G_c .

Agglomeration penalties

Finally, an important additional aspect of enforcement might be agglomeration penalties, i.e. larger penalties for larger cleared areas. Such targeting might occur in situations where enforcement is limited by the political feasibility of taking away subsistence livelihoods. This is likely in Thailand, which has a long history of tension between protected area regulations which prohibit all use and traditional land laws which encouraged "productive" land use (Fujita 2003, Giné 2005). Fujita (2003) notes that Royal Forestry Department officials were reluctant to

expropriate or destroy crops that had already been planted, even within protected forests. Albers and Grinspoon (1997) note that managers in Khao Yai National Park said they "rarely enforce regulations against collecting fruit, vegetables, grasses, and leaves in the park when the amounts collected imply that the goods are for family use rather than for sale."⁸ Although not officially sanctioned, it is likely that protected area managers, particularly in the national parks, might preferentially overlook small cleared patches which are perceived as subsistence use while enforcing prohibitions on larger patches of agricultural clearing.

An agglomeration penalty might be combined with any of the other spatial patterns of enforcement above. I model it in a simple fashion as an additional penalty for patches which are next to those already cleared at baseline before the protected area was established (e.g. Figure 1c above). For the case of a uniform penalty plus the agglomeration penalty, the parcel rents are:

(4)
$$E(R_i^a) = p^a f(q_i) - wl - t(d_i) - c - \pi_u G_u - I * G_{ap}$$

where *I* is an indicator function equal to one if a neighboring parcel was cleared at baseline and G_{ap} is the additional agglomeration penalty. This example is shown in Figure 2d.

3.3 Expected clearing and fragmentation as a function of enforcement type

Figure 2 shows representative patterns for each of the enforcement cases, following the equations given above (parameter values given at the bottom of Figure 2). While different parameter values produce different specific patterns, the equations above and the sample microlandscape can be used to gain a series of insights about how fragmentation measures tend to vary under different enforcement regimes. Table 1 summarizes these insights, first by comparing how different enforcement types will affect forest cover and forest patch size overall and in the interiors of parks (columns 1 and 2) and then by comparing how different types will affect forest cover (columns 3-6).

Compared to no protection, any enforcement will lower the rents on at least some parcels on the microlandscape, leading to less clearing and more forest cover (e.g. compare Figure 1d and Figure 2). In Table 1, this is noted in column 1 as (++) for all enforcement types. However, different enforcement patterns will differentially affect forest cover in the interior or core areas of the park versus at the periphery (Table 1, column 2). Spatially uniform or core-focused

⁸ Apparently villagers responded by collecting less per trip to remain below the limits!

enforcement will most strongly protect core areas (++ and +++) while boundary-focused enforcement will be less effective at protecting forest cover in the interior (+). This is because higher transportation costs mean lower rents for interior parcels. Under uniform or core-focused enforcement, these interior parcels are the first set to become unprofitable and thus more likely to stay in forest. Under boundary-focused enforcement, the higher transportation costs for interior parcels are partially offset by lower expected penalties, so there is relatively more interior clearing.

The effects of an agglomeration penalty are considered as an addition to the other types of enforcement (Table 1). Compared to no additional enforcement, an agglomeration penalty will add to forest protection overall (++). This penalty will also increase forest protection in the interiors of parks compared to no protection, but will be relatively less effective at protecting interior vs. exterior areas of parks (+). This is because large clearings, which attract the most punishment under the agglomeration penalty, are likely to be near the boundaries of protected areas, where transportation costs are low.⁹

Although this framework predicts differences across enforcement types, we still expect all types of enforcement to result in more forest cover than no protection, even in interior areas of parks, because enforcement lowers rents everywhere on the microlandscape. This prediction may not hold, however, if there is a high degree of labor immobility combined with boundaryfocused enforcement or enforcement with an agglomeration penalty. If labor cannot move easily across microlandscapes,¹⁰ effective protection in one part of the landscape will raise the profitability of parcels in other parts through decreased wages. Lower wages combined with low penalties in the interior could be enough to actually decrease interior forest cover compared to the no-protection case (-).

Different patterns of enforcement may also affect other measures of habitat fragmentation. Table 1 (columns 3-6) summarizes how different enforcement types are likely to

⁹ Note this assumes that transportation costs matter and are correlated with distance to the boundary of the protected area. If households are not well-integrated with markets (i.e. most agriculture is grown for household use and does not use market inputs), then clearing decisions would only be weakly related to transportation costs and the differences between enforcement types would be small.

¹⁰ There are likely to be at least some factors which limit the mobility of labor. In Thailand these include the system of identification and political registration which is tied to village of birth, poorly defined private property rights, and the advantages of extended family networks in home villages.

affect forest and cleared patch size, forest patch perimeter to area ratios and cleared patch density, conditional on forest cover.¹¹ The basic conclusion is that compared to a uniform enforcement pattern, we expect to see less fragmentation with core-focused enforcement and more with boundary-focused enforcement or enforcement with an agglomeration penalty (e.g. Figure 2). More specifically, core-focused enforcement (e.g. Figure 2c) tends to consolidate clearing closer to the edge of the microlandscape, preserving larger forest patches. This consolidation also tends to lead to fewer, larger cleared patches and smaller forest patch perimeter to area ratios. In contrast, boundary-focused enforcement (e.g. Figure 2b) tends to push more of the clearing into the interior of the microlandscape due to lower penalties in the interior. This is likely to split the forest into more small areas, reducing forest patch size, and to create a more branching and patchy pattern with many smaller cleared patches and high forest patch perimeter to area ratios.

Adding an agglomeration penalty to any of these enforcement patterns will also tend to increase fragmentation, conditional on forest cover (e.g. Figure 2d). The agglomeration penalty protects forest next to the areas already cleared while new clearing tends to spread across the microlandscape, resulting in smaller patch sizes, larger forest patch perimeter to area ratios and higher densities of cleared patches. Finally, it is important to consider how relaxing the exogenous wage assumption might affect these predictions. Adding labor immobility to any of the enforcement patterns will tend to spread clearing and increase fragmentation. Again, lower wages induced by the restrictions on clearing will reduce labor costs relative to transportation costs, increasing the relative profitability of parcels in the interior of the microlandscape and dispersing clearing.

In summary, this framework illustrates that policymakers who are concerned with patterns of forest clearing in addition to total amounts should carefully analyze the possible implications of different enforcement types. Uniform or core-focused enforcement is more likely to be effective at preventing interior clearing and forest fragmentation than boundary-focused enforcement or enforcement with agglomeration penalties. The next section explores whether

¹¹ Uniform enforcement generally produces similar patterns of fragmentation conditional on forest cover as a no protection case with lower agricultural prices. If production is linear in land quality, the patterns are the same (both a price decrease and a uniform penalty reduce rents by a constant amount); if production has diminishing returns in land quality then uniform enforcement tends to create somewhat more fragmentation.

Thai protected areas have prevented forest loss and fragmentation and how the observed patterns may reflect possible differences in enforcement types.

4. DATA AND EMPIRICAL STRATEGY

This section describes the empirical application of the microlandscapes approach. Regional landscapes in North and Northeast Thailand are divided into smaller units of analysis and fragmentation outcomes are assessed as a function of landscape and policy variables. In order to ensure that comparisons are made between units which are as similar as possible, I first pre-match the microlandscapes on the basis of characteristics affecting the location of protected areas that could also influence fragmentation, then estimate impacts using panel regression with microlandscape fixed-effects.

4.1 Data sources

The microlandscapes are defined using a 5 km x 5 km grid (and as a robustness check, a 10 km x 10 km grid). Figure 3 illustrates the scale of these microlandscapes compared to two national parks in Northern Thailand (Doi Inthanon and Ob Luang). Cleared areas and forested areas are shown in 1973 and 2000, giving an example of the increasing levels of deforestation and forest fragmentation seen over time both inside and outside of the protected areas.

For each microlandscape, I calculate the following dependent variables: forest cover, average forest patch size, maximum forest patch size, forest patch perimeter to area ratio, average cleared patch size and the density of cleared patches. These are calculated for 1973, 1985, 1992 and 2000, using land cover data which is summarized in Table 2. Land use for 1992 and 2000 is based on Landsat 5 Thematic Mapper (TM) images and for 1973 and 1985 on Landsat Multi-Spectral Scanner images (MSS). All layers contain information on cleared land, forested land, and water bodies, but only the 2000 layer distinguishes between different types of forest.¹² With respect to the resolution of the data, the pixel scale of the Landsat TM is 30 m x 30 m and of the Landsat MSS 60 m x 60 m. In comparison, village-level data from the Thai community development department NRD2C survey for Chiang Mai province in the 1990's

¹² The satellite-based classifications were compared to an analysis based on aerial photos and fieldwork that was done for a small number of sites in Northern Thailand at a similar time (Thomas et al. 2004). This indicated that the classified layers are fairly accurate at picking up fully cleared areas for growing rice or other annual crops, but that tree crops and early secondary regrowth (fallow) may be classified as forest.

reports approximately 1-2 rai of agricultural land per person on average, where one rai is equivalent to a 40 m x 40 m plot. This means that most clearing is likely to be picked up in this data, especially because individual landholdings tend to be larger in upland areas where most parks are located and because most clearing is done by multiple households working together.¹³

The decision to use a 5 km x 5 km grid to define the microlandscapes is admittedly somewhat arbitrary but was made to balance two important factors. On the one hand we want a relatively fine scale of resolution for the analysis (i.e. several microlandscapes within each park) and on the other we want a landscape unit that is large enough to matter ecologically. Median protected area size in the dataset was approximately 450 sq km with a minimum of approximately 50 sq km, so the 5 km x 5 km grid gives at least one and on average about 20 microlandscapes per park. This choice does limit the maximum patch size to be at most 25 sq km, but this is large enough to be ecologically relevant for key species. For instance, the white handed gibbon, which is an endangered forest canopy dweller in Thailand, has an estimated home range of $\sim 0.16-0.54$ sq km and is estimated to need 10 sq km for a genetically viable population.¹⁴ The average home range of the long-tailed Macqaw monkey, another relevant species, is approximately 1.25 sq km.¹⁵ However, the home ranges or areas needed for minimum populations for larger mammals (such as the Asian elephant) would be greater. As a robustness check to ensure that we do not miss important spatial patterns using small cells, the full dataset was re-created using a 10 km x 10 km grid, which increases the maximum forest patch size to 100 sq km.

4.2 Empirical strategy

As discussed in Section 2, land which has been selected for protected areas is usually more remote and rugged. More remote lands may be less likely to undergo fragmentation, because pressures to develop it are less; on the other hand more rugged lands might be more

¹³ Schmidt-Vogt (1998), for instance, documents areas equivalent to a square of 800m x 800m being cleared for one village. However, depending on terrain and ethnic group, individual households may also clear land (Schmidt-Vogt 1998) so it is possible that some very small clearings are not visible.

¹⁴ Wild animals rescue foundation of Thailand (www.warthai.org), "Gibbon Rehabilitation Project." Accessed October, 2011; Brockelman, W. & Geissmann, T. 2008. "Hylobates lar: IUCN Red List of Threatened Species. Accessed October 2011.

¹⁵ University of Michigan Museum of Zoology (http://animaldiversity.ummz.umich.edu) "Macaca fascicularis" Accessed October, 2011.

likely to undergo fragmentation because high quality land is very patchily distributed. Regardless of the expected direction, it is clear that in order to accurately assess the impacts of protected areas, we should identify effects based on a comparison to land with similar characteristics. Here, I first pre-process the data using matching to select a control group of microlandscapes which were never protected but have similar geographic characteristics to protected microlandscapes. I then estimate impacts using panel regression with microlandscape fixed effects, identifying effects by comparing changes over time between protected and unprotected landscapes.

Using matching to select control microlandscapes

To select a control group of microlandscapes which were never protected, I match on the basis of variables designed to capture factors which determined protected area locations and might also be correlated with deforestation or forest fragmentation.¹⁶ As discussed in further detail in Sims 2010 (Section 4.2), the particular history of protected areas in Thailand means that their locations were selected largely by central planners on the basis of observable physical and geographic characteristics. In particular, lands were more likely to be protected if they were important for watershed protection, had higher historical forest cover, included less high quality agricultural or timber land, were further from cities, protected scenic areas, and were closer to national borders. I therefore calculate and match on the following characteristics for each microlandscape: average elevation, average slope, maximum elevation, maximum slope, distance to major city (all of which were established before 1973), distance to railroad line, distance to Thai national boundary, distance to large river, distance to nearest major and minor road prior to protection (1962 roads), distance to nearest waterfall, and percent area in each major ecotype.¹⁷ Figure 4 shows the boundaries of the national parks and wildlife sanctuaries by period of implementation and illustrates that there is considerable spatial and temporal variation in protection.

The sources of data for the measures of protection and for each of the covariates are listed in Table 3. For the purposes of matching, protection must be a binary variable and is defined as

¹⁶ Matching methods help to select a control group for which there are high levels of overlap between covariates (Ho et al. 2007). This can reduce bias due to functional form choices in standard regression models in cases where treatment and overall control groups are quite dissimilar (Imbens 2007, Rosenbaum and Rubin 1985).

¹⁷ I use the logs of each of the slope, elevation and distance variables.

having any land in the 5 km x 5 km grid in a protected area by the year 2000. For the regression analysis, protection is defined continuously, as the share of each grid cell which is protected by 1973, 1985, 1992 or 2000. Table 4 gives summary statistics for the matched sample. The top half of Table 4 indicates that forest cover and average forest patch size have declined substantially across time, with the largest decrease in the earliest period (1973-1985). Median and maximum forest patch size also decline across time, while the average size of cleared patches increases. The average density of cleared patches in each microlandscape increases until 1992 and then decreases somewhat in the last time period, suggesting some consolidation of cleared areas in the 1990's.

I include in the sample available for matching all complete grid cells with more than 50% (majority) of forest cover in 1973,¹⁸ with less than 10% cloud cover in any image, and with less than 20% water in 1973. Matching is conducted separately for the North and Northeast regions, using one-to-one Mahalanobis matching with replacement. The summary statistics of the covariates for landscapes with and without protection (bottom half of Table 4) illustrate that matching does achieve substantial similarity across these covariates. After matching, only one of the normalized differences (distance to any road in 1962) is greater than .25 standard deviations, which is the rule of thumb for reasonable covariate balance suggested by Imbens and Wooldridge (2009).

Regression model: protected area impacts

Despite the matching, there may still be remaining differences between treated and control units in observables as well as potentially important unobservables such as the patchiness of land quality within a microlandscape.¹⁹ I therefore take advantage of the panel data in order to control directly for time-invariant unobservable characteristics by including microlandscape level fixed-effects. I estimate the following model:

(5)
$$\ln Y_{ijt} = \beta_1 WLS_{ijt} + \beta_2 NP_{ijt} + \alpha_i + \alpha_{jt} + \varepsilon_{ijt}$$

¹⁸ This focuses the analysis on landscapes which have significant forest assets to start and thus would be the subject of concern about forest fragmentation. However, the results are robust to including all grid cells with more than 25% or 75% forest cover in 1973 (results available from author).

¹⁹ For instance, Butler et al. (2004) finds that model fit improves when the spatial configuration of returns within each landscape is included.

where Y_{ijt} is the fragmentation outcome for microlandscape *i* in province *j* at time *t*. *WLS*_{ijt} and *NP*_{ijt} are the share of microlandscape *i* in province *j* which was protected at time *t* in wildlife sanctuaries and national parks. The microlandscape fixed effect is denoted α_i . I also include a set of province-year fixed effects (denoted as α_{jt}) to control for regional trends.²⁰ All specifications are run with robust standard errors, clustered at the level of the district (*amphoe*) to account for possible spatial and serial autocorrelation.²¹

5. RESULTS AND DISCUSSION

Following the predictions from the economic framework outlined in Section 3, I first discuss estimated impacts of protection on forest cover and forest patch size, overall and in the interiors of protected areas. I then discuss impacts on fragmentation metrics, conditional on forest cover.

5.1 Effects of protection on forest cover and forest patch size

Table 5 shows results from the fixed effects regression model (Equation 5) with forest cover and forest patch size as outcomes. Columns 1, 3, and 5 give the average impact for protected areas and columns 2, 4, and 6 allow effects to vary across wildlife sanctuaries and national parks. The results show, first, that protection did result in significant increases in forest cover. To interpret the magnitudes, recall that the regressions use the logs of the dependent variables and that the key policy variable is the share of each landscape protected. Therefore the coefficients indicate the approximate percentage change in the dependent variables corresponding to a change in the share of protection from 0 to 1 (i.e. a change from none of the microlandscape protected to all of it protected). Forest cover is thus estimated to be ~19% higher overall due to protection, or ~22% higher due to protection as a wildlife sanctuary and ~ 17% higher due to protection as a national park (columns 1 and 2). This suggests a possibly higher level of enforcement in the wildlife sanctuaries, but the difference is not significant (p-values for F-tests of equality of coefficients are in the last row).

²⁰ I check the robustness of the results to using lagged protection variables as well. The increases in forest cover and forest patch size are robust to using protection from the previous period and to including both current and lagged protection variables (results available from author).

²¹ As an alternate robustness check to deal with possible spatial autocorrelation, I also take a random 20% sample of the dataset. The significant decreases in forest fragmentation are robust to this check (results available from author).

Second, Table 5 indicates that protected areas increased average and maximum forest patch size, by approximately 25% and 21% respectively. Wildlife sanctuary designation resulted in a more than 30% increase in average forest patch size, while national park designation increased average forest patch size by approximately 20%, although the difference between the two is again not statistically significant. Given that the mean of average forest patch size in the matched controls in 2000 was 10.3 sq km, a 20% increase represents an addition of approximately 2 sq km of forest to average patch size, while a 30% increase represents an addition of 3 sq km of forest. Maximum patch size increased by approximately 24% for the wildlife sanctuaries and approximately 19% for the national parks. As maximum patch size was on average 16.2 sq km in the matched controls, a 20% increase corresponds to a more than 3 sq km increase. These changes therefore represent potentially ecologically meaningful differences, given the home range sizes for the two primate species discussed previously. Table 6 shows the same set of regressions for the 10 km x 10 km microlandscapes, which allow for larger patch sizes. The results are similar, although the estimated percentage increases in average forest patch size and maximum forest patch size are higher. There is also a significant difference in the effect on average forest patch size between wildlife sanctuaries and national parks, with a greater impact on forest patch size due to wildlife sanctuary protection (column 4).

5.2 Effects of protection on interior vs. periphery areas

A key prediction of the economic framework described in Section 3 was that boundaryfocused enforcement or enforcement with an agglomeration penalty would be less effective at protecting forest cover in the interior areas of parks. The summary statistics (Table 4) indicate potential differences between wildlife sanctuaries and national parks. When we divide protected microlandscapes into those which are fully in the interior of protected areas and those which overlap with protected area boundaries, we see that forest cover in the year 2000 is indeed somewhat higher in the interiors of wildlife sanctuaries (94.7%) versus the interiors of national parks (92.4%) and the difference in means (2.3%) is significant at the 5% level. However, forest cover is also higher at the exteriors of wildlife sanctuaries than national parks (1.8%).

To more rigorously test how protection has affected forest cover and forest patch size in the interiors of protected areas, I re-run the regression specification given in Equation 5 above, but allow the marginal effects of protection to vary flexibly with protected area type and distance

to protected area boundaries.²² Figure 5 graphs these marginal effects. The top panel shows results for forest cover outcomes and the bottom panel for average forest patch size. Although the confidence intervals overlap, different patterns are evident between wildlife sanctuaries and national parks. The wildlife sanctuaries are more effective overall at protecting forest cover and increasing forest patch size, and this effectiveness is maintained even in core areas. National parks, by contrast, show a downward trend in effectiveness for forest cover in interior areas and appear to have actually decreased average forest patch size in interior areas, compared to no protection. Interestingly, national parks also appear to be slightly less effective close to park boundaries, suggestive of either the agglomeration penalties hypothesis or possibly a tacit acceptance by authorities of some encroachment near the boundaries.

In summary, although both national parks and wildlife sanctuaries were effective in increasing forest cover and forest patch size, differential effectiveness in the interior versus periphery areas is consistent with spatial patterns of uniform or core-focused enforcement in the wildlife sanctuaries and boundary-focused enforcement or enforcement with agglomeration penalties in the national parks. Yet the results also suggest a possibly greater level of enforcement in the wildlife sanctuaries. To further explore differences between the two, it is useful to consider impacts on additional fragmentation outcomes while controlling for the total amount of forest cover.

5.3 Effects of protection on fragmentation, conditional on forest cover

Table 7 shows protected area effects on forest patch size, cleared patch size, forest patch perimeter to area ratios, and cleared patch density, conditional on forest cover. The results indicate that protected areas did not significantly change average and maximum forest patch sizes, conditional on forest cover (column 1). This suggests that much of the unconditional increase in forest patch size can be explained by an increase in the total amount of forest cover rather than changes in patterns of clearing. Wildlife sanctuaries did, however, significantly increase average cleared patch size and significantly reduce forest patch perimeter to area ratios, conditional on forest cover (columns 3 and 4). This suggests a possible reduction in fragmentation through consolidation of agricultural patches in the wildlife sanctuaries. In

²² Regressions are the same as in Equation 5 and Table 5 but include interactions between the share protected as national park or as wildlife sanctuary and distance, distance squared and distance cubed.

contrast, national parks significantly increased the density of cleared patches (column 5) and may have increased forest patch perimeter to area ratios (column 4), suggesting increased fragmentation through dispersion of clearing.

Comparing the two types of protected areas to each other, the results indicate that wildlife sanctuaries resulted in significantly less fragmentation, conditional on forest cover, than national parks (p-values for F-tests of equality of coefficients in the last row of Table 7). Wildlife sanctuaries retained marginally significantly larger average forest patch sizes (p-value = .084), significantly larger cleared patch sizes (p=.005), significantly smaller forest patch perimeter to area ratios (p=.020) and significantly lower density of cleared patches (p=.002) than national parks. As a robustness check, Table 8 shows the same regressions for the 10 km x 10 km grid. Note that the signs of the coefficients are the same in all cases as in Table 7. With the larger microlandscapes, wildlife sanctuaries retained significantly larger forest patch sizes (p=.042), significantly larger average cleared patch sizes (p=.009) and significantly smaller cleared patch density (p=.003) than national parks. Forest patch perimeter to area ratios are also estimated to be smaller in wildlife sanctuaries, although the difference is no longer statistically significant (p=0.137).

Taken together, the estimates of protected area effects on fragmentation conditional on forest cover indicate that wildlife sanctuaries resulted in significantly less fragmentation than national parks. Recalling the predictions from the economic framework (Table 1), these differences in patterns are consistent with spatial patterns of enforcement which are uniform or core-focused in the wildlife sanctuaries and boundary-focused enforcement approach or include agglomeration penalties in the national parks.

One potential concern is that these differences could be driven by differences in location rather than in enforcement patterns. As is shown in Figure 4, wildlife sanctuaries and national parks often share borders and were located in generally similar areas, but some differences remain.²³ Given the inclusion of microlandscape fixed-effects, the observed differences in fragmentation outcomes are not likely to be driven by fixed differences in landscape characteristics, but they could be driven by differential changes in access to transportation across

²³ Using the rule of thumb of .25 standard deviations as a substantial difference, landscapes with wildlife sanctuaries were sited further from 1962 roads and railways and had less ecoregion 3 forest type (tropical and sub-tropical dry broadleaf forest).

time. Two relevant robustness checks were performed: matching was conducting adding distance to late 1990's roads instead of and in addition to 1963 roads as a covariate, and the regressions were repeated including district-year fixed effects rather than province-year fixed effects to better control for possible local changes in access to transportation. The results are robust to both of these changes (available on request).

6. CONCLUSION

Functioning ecosystems provide goods and services that are valued directly or indirectly by humans, such as water supplies, plant products, wild game, and species biodiversity (Daily 1997, Boyd and Banzhaf 2007). Since the productivity of ecosystems depends on their spatial configuration in addition to overall quantities, resource managers need to understand how landuse policies will influence habitat fragmentation. Retrospective studies of past conservation policies can provide useful information to help understand the possible ramifications of current policy decisions. This paper has illustrated a method for retrospective analysis of fragmentation effects using program evaluation methods in combination with common fragmentation statistics. The method was applied to analyze how protected areas in Thailand affected forest fragmentation in the nearly 30 years between 1973 and 2000.

The results indicate that that protected areas in North and Northeast Thailand did significantly increase forest cover and forest patch size compared to the counterfactual of no protection. This is encouraging as it suggests these were not purely "paper parks"; they did make a real difference in preserving contiguous areas of forest habitat. However, differences in forest cover and forest patch size in the core vs. periphery areas of parks and in fragmentation metrics conditional on forest cover suggest different impacts of wildlife sanctuary and national park protection on spatial patterns of forest. Wildlife sanctuaries were more effective at maintaining forest cover and forest patch size in interior areas and conditional on the level of forest cover, resulted in marginally significantly larger forest patch sizes, significantly larger cleared patch sizes, significantly smaller forest patch perimeter to area ratios and significantly lower densities of cleared patches than national parks.

The differences in fragmentation outcomes between wildlife sanctuaries and national parks indicate possible differences in the spatial patterns of enforcement across protected area types. Given the lack of direct measures of enforcement, the empirical results are linked to

possible enforcement patterns based on their consistency with predictions from a simple von Thünen style economic framework. The results are consistent with a uniform or core-focused spatial enforcement approach in the wildlife sanctuaries and a boundary-focused approach or enforcement with agglomeration penalties in the national parks. Despite the lack of direct evidence on enforcement, these results are important as they highlight that different types of protection may result in significantly different patterns of land use. Since these differences are potentially driven by different spatial patterns of enforcement, they reinforce existing theoretical work urging conservation managers to consider how enforcement affects both total deforestation and patterns of deforestation when allocating scarce enforcement resources.

The gap in data on enforcement patterns is one that holds globally and is due to the expense and political difficulties of collecting such information. Clearly, this type of data collection is a priority for future work on protected area effectiveness. Future work should also consider how protected area enforcement might be better structured to limit economic costs and maximize environmental benefits. Protected areas in Thailand have historically been managed under an awkward combination of legal prohibition of agricultural use and *de facto* recognition of community livelihoods. This gap between the legal system and day-to-day management and enforcement has caused considerable controversy and conflict (Vandergeest 1996, Albers and Grinspoon 1997, Laungaramsri 2000, ICEM 2003, Chettamart 2003, Emphandhu and Chettamart 2003, Roth 2004a,b, Delcore 2007). If, as the results for national parks suggest, incomplete enforcement which tries to protect boundaries or overlooks subsistence use has led to more clearing of small patches or more interior clearing, this is doubly unfortunate. Such clearing patterns likely indicate lower economic returns in addition to higher fragmentation. These results underscore previous arguments (Albers and Grinspoon 1997, Albers and Muller 2004, Albers 2010) that protected area managers should be given more flexibility in restrictions across space as well as more resources. Finally, future work should also use similar methods to determine whether the results hold for other countries. Thailand is a middle-income country which enjoyed relative stability throughout this period. Many developing countries likely have protected areas with even fewer enforcement resources available and higher potential for perverse fragmentation effects. A better understanding of empirical results across multiple cases could hopefully lead to enforcement patterns which simultaneously improve protection of ecosystems and local livelihoods.

REFERENCES

- Albers, H. J. (1996). "Modeling Ecological Constraints on Tropical Forest Management: Spatial Interdependence, Irreversibility, and Uncertainty." Journal of Environmental Economics and Management 30(1): 73-94.
- Albers, H. J. (2001). A Spatial-Intertemporal Model for Tropical Forest Management Applied to Khao Yai National Park, Thailand, RFF Discussion Paper 01-35.
- Albers, H. J. (2010). "Spatial modeling of extraction and enforcement in developing country protected areas." Resource and Energy Economics 32(2): 165-179.
- Albers, H. J., A. W. Ando and X. Chen (2008). "Spatial-econometric analysis of attraction and repulsion of private conservation by public reserves." Journal of Environmental Economics and Management 56(1): 33-49.
- Albers, H. J., A. W. Ando and J. Shogren (2010). "Introduction to spatial natural resource and environmental economics." Resource and Energy Economics 32(2): 93-97.
- Albers, H.J., A.W. Ando, M. Bu, and M.G. Wing. (2012). "Road-network agglomeration, road density, and protected-area fragmentation." Letters in Spatial and Resources Sciences 5: 137-150.
- Albers, H. J. and M. Bu (2009). Economics of Habitat Fragmentation and Fragmentation Policy: A Literature Review, Oregon State Applied Economics/ Forest Resources.
- Albers, H. J. and E. Grinspoon (1997). "A comparison of the enforcement of access restrictions between Xishuangbanna Nature Reserve (China) and Khao Yai National Park (Thailand)." Environmental Conservation 24(4): 351-362.
- Albers, H. J. and J. Muller (2004). "Enforcement, payments, and development projects near protected areas: how the market setting determines what works where." Resource and Energy Economics 26: 185-204.
- Alig, R. J., D. J. Lewis and J. J. Swenson (2005). "Is forest fragmentation driven by the spatial configuration of land quality?: The case of western Oregon." Forest Ecology and Management 217(2-3): 266-274.
- Allan, B. F., F. Keesing and R. S. Ostfeld (2003). "Effect of forest fragmentation on Lyme disease risk." Conservation Biology 17: 267-72.
- Andam, K. S., P. J. Ferraro, A. Pfaff, A. Sanchez-Azofeifa and J. Robalino (2008). "Measuring the effectiveness of protected area networks in reducing deforestation." Proceedings of the National Academy of Sciences 105(42).
- Andam, K. S., P. J. Ferraro, K. R. E. Sims, A. Healy and M. B. Holland (2010). "Protected areas reduced poverty in Costa Rica and Thailand." Proceedings of the National Academy of Sciences 107(22): 9996-10001.
- Ando, A., J. Camm, S. Polasky and A. Solow (1998). "Species Distributions, Land Values, and Efficient Conservation " Science 279(5359).
- Ando, A. W. and P. Shah (2010). "Demand-side factors in optimal land conservation choice." Resource and Energy Economics 32(2): 203-221.

- Angelsen, A. (2007). Forest Cover Change in Space and Time: Combining the von Thünen and Forest Transition Theories, World Bank Policy Research Working Paper 4117.
- Angelsen, A. (2010). "Policies for reduced deforestation and their impact on agricultural production." Proceedings of the National Academy of Sciences 107(46): 19639-19644.
- Armsworth, P. R., B. E. Kendall and F. W. Davis (2004). "An Introduction to Biodiversity Concepts for Environmental Economists." Resource and Energy Economics 26: 115-136.
- Bender, D. J., T. A. Contreras and L. Fahrig (1998). "Habitat loss and population decline: a metaanalysis of the patch size effect." Ecology 79: 517-33.
- Betts, M. (2000). "In Search of Ecological Relevancy: A Review of Landscape Fragmentation Metrics and their Application for the Fundy Model Forest." University of New Brunswick, 38 pp.
- Bockstael, N. E. (1996). "Modeling Economics and Ecology: The Importance of a Spatial Perspective." American Journal of Agricultural Economics 78(5): 1168-1180.
- Boose, E. R., D. R. Foster and M. Fluet (1994). "Hurricane impacts to tropical and temperate forest landscapes." Ecological Monographs 64: 369-400.
- Boyd, J. and S. Banzhaf (2007). "What are ecosystem services? The need for standardized environmental accounting units " Ecological Economics 63(2-3): 616-626
- Butler, B. J., J. J. Swenson and R. J. Alig (2004). "Forest fragmentation in the Pacific Northwest: quantification and correlations." Forest Ecology and Management 189(1-3): 363-373.
- Chettamart, S. (2003). Ecotourism Resources and Management in Thailand. Paper presented at Malaysia-Thailand Technology and Business Partnership Dialogue, 27-28 July, Langkawi, Malaysia.
- Chomitz, K. M. and D. A. Gray (1996). "Roads, Land Use, and Deforestation: A Spatial Model Applied to Belize." World Bank Economic Review 10(3): 487-512.
- Chape, S., J. Harrison, M. Spalding, and I. Lysenko (2005). "Measuring the extent and effectiveness of protected areas as an indicator for meeting global biodiversity targets." *Philosophical Transactions of the Royal Society Series B*, 360: 443-455.
- Christensen, N. L. (1997). Managing for Heterogeneity and Complexity on Dynamic Landscapes. The Ecological Basis of Conservation: Heterogeneity, Ecosystems, and Biodiversity. S. T. Pickett, R. S. Ostfeld, M. Shachak and G. E. Likens, Chapman and Hall.
- Costello, C., S. D. Gaines and J. Lynham (2008). "Can Catch Shares Prevent Fisheries Collapse?" Science 321(5896): 1678-1681.
- Cropper, M., C. Griffiths and M. Mani (1999). "Roads, Population Pressures, and Deforestation in Thailand 1976-1989." Land Economics 75(1): 58-73.
- Cropper, M., J. Puri and C. Griffiths (2001). "Predicting the Location of Deforestation: The Role of Roads and Protected Areas in North Thailand." Land Economics 77(2): 172-186.
- Daily, G. C., Ed. (1997). Nature's Services: Societal Dependence on Natural Ecosystems. Washington, D.C., Island Press.

- Deininger, K. and B. Minten (2002). "Determinants of Deforestation and the Economics of Protection: An Application to Mexico." American Journal of Agricultural Economics 84(4): 943-960.
- Delcore, H. D. (2007). "The Racial Distribution of Privilege in a Thai National Park." Journal of Southeast Asian Studies 38(1): 83-105.
- Dixon, J. A. and P. B. Sherman (1990). Economics of Protected Areas. Washington, DC, Island Press.
- Emphandhu, D. and S. Chettamart (2003). Thailand's Experience in Protected Area Management. Bangkok, Thailand, Faculty of Forestry, Kasetsart University.
- Fahrig, L. (2002). "Effect of habitat fragmentation on the extinction threshold: A synthesis." Ecological Applications 12(2): 346-353.
- Ferraro, P. (2008). Protected Areas and Human Well-Being: Economics and Conservation in the Tropics: A Strategic Dialogue, RFF Conference Paper.
- Ferraro, P. J., M. M. Hanauer and K. R. E. Sims (2011). "Conditions associated with protected area success in conservation and poverty reduction." Proceedings of the National Academy of Sciences 108(34): 13913-13918.
- Ferraro, P. J., C. McIntosh and M. Ospina (2007). "The effectiveness of the US endangered species act: An econometric analysis using matching methods." Journal of Environmental Economics and Management 54(3): 245-261.
- Fischer, J. and D. B. Lindenmayer (2007). "Landscape modification and habitat fragmentation: a synthesis." Global Ecology and Biogeography 16(3): 265-280.
- FAO Regional Office for Asia and the Pacific (2009). "Asia-Pacific Forestry Sector Outlook Study II: Thailand Forestry Outlook" Working Paper No. APFSOS II/WP/2009/22, Bangkok, Thailand.
- Fox, J., J. Krummel, S. Yarnasarn, M. Ekasingh and N. Podger (1995). "Land use and landscape dynamics in northern Thailand: Assessing change in three upland watersheds." AMBIO 24(6): 328-334.
- Fox, J. and J. B. Vogler (2005). "Land-Use and Land-Cover Change in Montane Mainland Southeast Asia " Environmental Management 36(3): 2005.
- Fujita, W. (2003). "Dealing with Contradictions: Examining National Forest Reserves in Thailand." Southeast Asia Studies 41(2): 206-238.
- Galvin, K. A., R. S. Reid, R. H. Behnke Jr. and N. T. Hobbs, Eds. (2008). Fragmentation in Semi-Arid and Arid Landscapes: Consequences for Human and Natural Systems, Springer Netherlands.
- Ganjanapan, A., Ed. (2000). Local Control of Land and Forests: Cultural Dimensions of Resource Management in Northern Thailand. Chiang Mai, Regional Center for Social Science and Sustainable Development.
- Giné, X. (2005). Cultivate or Rent out? Land Security in Rural Thailand, Working Paper: World Bank.

- Greenstone, M. and T. Gayer (2009). "Quasi-experimental and experimental approaches to environmental economics." Journal of Environmental Economics and Management 57(1): 21-44.
- Gustafson, E. J. (1998). "Quantifying landscape spatial pattern: What is the state of the art." Ecosystems: 143-156.
- Hawbaker, T., V. Radeloff, R. Hammer and M. Clayton (2005). "Road Density and Landscape Pattern in Relation to Housing Density, and Ownership, Land Cover, and Soils." Landscape Ecology 20(5): 609-625.
- Hawbaker, T. J., V. C. Radeloff, M. K. Clayton, R. B. Hammer and C. E. Gonzalez-Abraham (2006). "Road Development, Housing Growth and Landscape Fragmentation in Northern Wisconsin: 1937-1999." Ecological Applications 16(3): 1222-1237.
- Heilman, G. E., J. R. Strittholt, N. C. Slosser and D. A. Dellasala (2002). "Forest Fragmentation of the Conterminous United States: Assessing Forest Intactness through Road Density and Spatial Characteristics." BioScience 52(5): 411-422.
- Hirsch, P. and L. Lohmann (1989). "Contemporary politics of environment in Thailand." Asian Survey 29(4)(439-453).
- Ho, D. E., K. Imai, G. King and E. A. Stuart (2007). "Matching as Nonparametric Preprocessing for Reducing Model Dependence in Parametric Causal Inference." Political Analysis 15(3): 199-236.
- Horan, R. D., J. F. Shogren and B. M. Gramig (2008). "Wildlife conservation payments to address habitat fragmentation and disease risks." Environment and Development Economics 13(03): 415-439.
- Imbens, G. (2007). Estimation of Average Treatment Effects Under Unconfoundedness Cambridge, MA, NBER Summer Institute, "What's New in Econometrics, Lecture 1".
- Imbens, G. and J. Wooldridge (2009). "Recent Developments in the Econometrics of Program Evaluation." Journal of Economic Literature 47 (1): 5-86.
- International Center for Environmental Management (ICEM) (2003). Thailand National Report on Protected Areas and Development. Review of Protected Areas and Development in the Lower Mekong River Region. Indooroopilly, Queensland, Australia.
- Kramer, M. G., A. J. Hansen, M. L. Taper and E. J. Kissinger (2001). "Abiotic controls on longterm windthrow disturbance and temperate rain forest dynamics in southeast Alaska." Ecology 82(2749–68).
- Laungaramsri, P. (2000). "The Ambiguity of "Watershed": The Politics of People and Conservation in Northern Thailand." SOJOURN: Journal of Social Issues in Southeast Asia 15: 52–75.
- Leitao, A. B. and J. Ahern (2002). "Applying landscape ecological concepts and metrics in sustainable landscape planning." Landscape and Urban Planning 59(2): 65-93.
- Lewis, D. J. (2010). "An economic framework for forecasting land-use and ecosystem change." Resource and Energy Economics 32(2): 98-116.

- Lewis, D. J. and R. J. Alig (2009). "Empirical Methods for Modeling Landscape Change, Ecosystem Services and Biodiversity." Western Economics Forum 8(1): 29-39.
- Lewis, D. J., G. L. Hunt and A. J. Plantinga (2002). "Public Conservation Land and Employment Growth in the Northern Forest Region." Land Economics 78(2): 245-259.
- Lewis, D. J., G. L. Hunt and A. J. Plantinga (2003). "Does Public Lands Policy Affect Local Wage Growth?" Growth and Change 34(1): 64-86.
- Lewis, D. J. and A. J. Plantinga (2007). "Policies for Habitat Fragmentation: Combining Econometrics with GIS-Based Landscape Simulations." Land Economics 83(2): 109-127.
- Lewis, D. J., A. J. Plantinga, E. Nelson and S. Polasky (2011). "The efficiency of voluntary incentive policies for preventing biodiversity loss." Resource and Energy Economics 33(1): 192-211.
- Lewis, D. J., A. J. Plantinga and J. Wu (2009). "Targeting Incentives to Reduce Habitat Fragmentation." American Journal of Agricultural Economics 91(4): 1080-1096.
- Lewis, D. J. and J. Wu (2005). Optimal Landscape Conservation with Habitat Fragmentation Effects. Working Paper, Department of Agricultural and Applied Economics, University of Wisconsin-Madison.
- Liu, J., M. Linderman, Z. Ouyang, L. An, J. Yang and H. Zhang (2001). "Ecological Degradation in Protected Areas: The Case of Wolong Nature Reserve for Giant Pandas." Science 292(5514): 98-101.
- McGarigal, K. and S. A. Cushman (2002). "Comparative evaluation of experimental approaches to the study of habitat fragmentation effects." Ecological Applications 12: 335-45.
- Ochoa-Gaona, S. (2001). "Traditional Land-Use Systems and Patterns of Forest Fragmentation in the Highlands of Chiapas, Mexico." Environmental Management 27(4): 571-586.
- O'Neill, R. V. and et al. (1988). "Indices of landscape pattern." Landscape Ecology 1: 153-162.
- Opdam, P. and D. Wascher (2004). "Climate change meets habitat fragmentation: linking landscape and biogeographical scale levels in research and conservation "Biological Conservation 117(3): 285-297.
- Pfaff, A., J. Robalino, G. A. Sanchez-Azofeifa, K. S. Andam and P. Ferraro (2009). "Park Location Affects Forest Protection: Land Characteristics Cause Differences in Park Impacts across Costa Rica." B.E. Journal of Economic Analysis and Policy 9(2).
- Pfaff, A., J. Robalino, R. Walker, S. Aldrich, M. Caldas, E. Reis, S. Perz, C. Bohrer, E. Arima, W. Laurance and K. Kirby (2007). "Road Investments, Spatial Spillovers, and Deforestation in the Brazilian Amazon." Journal of Regional Science 47(1): 109-123.
- Pfaff, A., J. Robalino, R. Walker, S. Perz, W. Laurance, C. Bohrer, S. Aldrich, E. Arima, M. Caldas and K. Kirby (2011). Heterogeneous Transport Impacts and 'Clean' Development: nonmono-Thunic deforestation from roads in the Brazilian Amazon. Working paper, Duke University: 109-123.
- Pfaff, A. S. P. (1999). "What Drives Deforestation in the Brazilian Amazon? Evidence from Satellite and Socioeconomic Data." Journal of Environmental Economics and Management 37: 26-43.

- Riitters, K. H., J. D. Wickham, R. V. O'Neill, K. B. Jones, E. R. Smith, J. W. Coulston, T. G. Wade and J. H. Smith (2002). "Fragmentation of Continental United States Forests." Ecosystems 5: 815-822.
- Robalino, J. A. (2007). "Land Conservation Policies and Income Distribution: Who Bears the Burden of our Environmental Efforts?" Environment and Development Economics 12(4): 521-533.
- Robalino, J. and A. Pfaff (2012). "Contagious Development: Neighbor Interactions in Deforestation." Journal of Development Economics 97(2): 427-436.
- Robinson, E., J. Williams and H. Albers (2002). "The Influence of Markets and Policy on Spatial Patterns of Non-Timber Forest Product Extraction." Land Economics 78(2): 260-271.
- Robinson, E. J. Z., H. J. Albers and J. C. Williams (2005). Analyzing the Impact of Excluding Rural People from Protected Forests: Spatial Resource Degradation and Rural Welfare, The Centre for the Study of African Economies Working Paper Series.
- Robinson, E. J. Z., H. J. Albers and J. C. Williams (2008). "Spatial and Temporal Aspects of Non-Timber Forest Product Extraction: The Role of Community Resource Management." Journal of Environmental Economics and Management 56(3): 234-245.
- Robinson, E. J. Z., H. J. Albers and J. C. Williams (2011). "Sizing Reserves within a Landscape: The Roles of Villagersâ€^M Reactions and the Ecological-Socioeconomic Setting." Land Economics 87(2): 233-249.
- Robinson, E. J. Z. and R. B. Lokina (2011). "A spatial-temporal analysis of the impact of access restrictions on forest landscapes and household welfare in Tanzania." Forest Policy and Economics 13(1): 79-85.
- Rochelle, J. A., L. Lehmann and J. Wisniewski (1999). Forest Fragmentation: Wildlife and Management Implications Boston, Brill.
- Rosenbaum, P. R. and D. B. Rubin (1985). "Constructing a Control Group Using Multivariate Matched Sampling Methods that Incorporate the Propensity Score." The American Statistician 39(1): 33-38.
- Roth, R. (2004a). "On the Colonial Margins and in the Global Hotspot: Park-People Conflicts in Highland Thailand." Asia Pacific Viewpoint 45(1): 13-32.
- Roth, R. (2004b). "Spatial Organization of Environmental Knowledge: Conservation Conflicts in the Inhabited Forest of Northern Thailand." Ecology and Society 9(3).
- Ruhle, G. (1964). Advisory Report on a National Park System for Thailand, 1959-1960, A Report Prepared for the International Union for Conservation of Nature and Natural Resources and the American Committee for International Wild Life Protection, New York.
- Samek, J. H., D. X. Lan, C. Silapathong, C. Navanagruha, S. M. S. Abdullah, I. Gunawan, B. Crisostomo, F. Hilario, H. M. Hien, D. L. Skole, W. Chomentowski, W. A. Salas and H. Sanjaya (2004). Land Use and Land Cover Change in Southeast Asia. Land Change Science: Observing, Monitoring and Understanding Trajectories of Change on the Earth's Surface. G. Gutman and et al. Netherlands, Springer.

- Samuelson, P. A. (1983). "Thünen at Two Hundred." Journal of Economic Literature 21(4): 1468-1488.
- Sánchez-Azofeifa, G., G. C. Daily, A. S. P. Pfaff and C. Busch (2003). "Integrity and Isolation of Costa Rica's National Parks and Biological Reserves: Examining the Dynamics of Landcover Change." Biological Conservation 109(1): 123-135.
- Sánchez-Azofeifa, G., C. Quesada-Mateo, P. Gonzalez-Quesada, S. Dayanandan and K. S. Bawa (1999). "Protected Areas and Conservation of Biodiversity in the Tropics." Conservation Biology 13(2): 407-411.
- Sanchez-Azofeifa, G. A., R. C. Harriss and D. L. Skole (2001). "Deforestation in Costa Rica: A Quantitative Analysis Using RemoteSensing Imagery." Biotropica 33(3): 378-384.
- Schmidt-Vogt, D. (1998). "Defining degradation: the impacts of swidden on forests in northern Thailand." Mountain Research and Development 18(2): 135-149.
- Sims, K. R. E. (2010). "Conservation and development: Evidence from Thai protected areas." Journal of Environmental Economics and Management 60(2): 94-114.
- Souris, M. (2007). Thailand GIS Data, http://www.rsgis.ait.ac.th/~souris/thailand.htm.
- Stanfield, B. J., J. C. Bliss and T. A. Spies (2002). "Land ownership and landscape structure: a spatial analysis of sixty-six Oregon (USA) Coast Range watersheds." Landscape Ecology 17(8): 685-697.
- Stavins, R. N. and A. B. Jaffe (1990). "Unintended Impacts of Public Investments on Private Decisions: The Depletion of Forested Wetlands." American Economic Review 80(3): 337-352.
- Thomas, D. E. (1996). Forests for a Dynamic Kingdom: Support for the Emergence of Social Forestry in Thailand. Chiang Mai, Thailand, Ford Foundation Thailand Social Forestry Program.
- Thomas, D. E., P. Preechapanya and P. Saipothong (2004). Developing Science-Based Tools for Participatory Watershed Management in Montane Mainland Southeast Asia: Final Report to the Rockefeller Foundation. Chiang Mai, Thailand, International Centre for Research on Agroforestry (ICRAF).
- Turner, I. M. (1996). "Species Loss in Fragments of Tropical Rain Forest: A Review of the Evidence." Journal of Applied Ecology 33(2): 200-209.
- Turner, M. G. (1990). "Spatial and temporal analysis of landscape patterns." Landscape Ecology 4: 21-30.
- Turner, M. G. (2005). "Landscape Ecology: What Is the State of the Science?" Annual Review of Ecology, Evolution, and Systematics 36: 319-344.
- Urban, D. L., R. V. O'Neill and J. H. H. Shugart (1987). "Landscape ecology: A hierarchical perspective can help scientists understand spatial patterns." BioScience 37: 119-127.
- Vandergeest, P. (1996). "Property rights in protected areas: Obstacles to community involvement as a solution in Thailand." Environmental Conservation 23(3): 259-268.
- Vandergeest, P. (2003). "Racialization and Citizenship in Thai Forest Politics." Society and Natural Resources 16: 19-37.

- Viña A., F.R. Echavarria and D.C. Rundquist (2004). "Satellite Change Detection Analysis of Deforestation Rates and Patterns along the Colombia Ecuador Border." AMBIO 33(3):118-125.
- Zimmerer, K. S., R. E. Galt and M. V. Buck (2004). "Globalization and Multi-spatial Trends in the Coverage of Protected-Area Conservation (1980-2000)." AMBIO 33(8): 520–529

Figure 1: Economic framework for clearing on a sample microlandscape

Land quality (a) and distance to market (b) provide underlying variation in rents which drive clearing at baseline (c) and as agricultural prices increase (d).



c) Rents and baseline clearing (p^a =.85, 10% cleared)

The sample microlandscape consists of 30 x 30 parcels. Solid blue parcels indicate water, red square at bottom indicates a road leading to the closest market. Panel a) indicates land quality as a function of parcel distance to water source $(q_i = 10 - dw_i)$; darker shading corresponds to higher quality. Panel b) indicates distance to market, d_i ; darker shading corresponds to closer to market. Panel c) indicates parcel rents as determined by Equation 1 with agricultural price, $p^a = .85$ and parameter values: $f(q_i) = 3.8(q_i)$, l=1, w=5, t=.45, c=15. White parcels are cleared (rents >= 0) while green remain in forest (rents < 0). 10% of the landscape is cleared. Panel d) indicates parcel rents and clearing after an increase in agricultural prices ($p^a=1.11, 45\%$ cleared).

Figure 2: Clearing and fragmentation under different enforcement patterns

Same sample landscape as Fig. 1 with protected area established before the increase in ag. prices.



Microlandscape characteristics are the same as in Figure 1, but the landscape is now part of a protected area with the boundary at the bottom of the microlandscape (dotted black line). Parcel rents are determined by Equations 2-4. a) Uniform enforcement, with $\pi_u G_u = 5.5$. b) Boundary-focused enforcement with $\pi_b (b_i) G_b = 1.4 + 0.26(31 - b_i)$ where b_i is parcel distance from the park boundary. c) Core-focused enforcement with $\pi_c(b_i)G_c = 14.7-0.5(31-b_i)$. d) Uniform enforcement + agglomeration penalty with $\pi_u G_u = 3.6$ and $G_{ap}=10.4$. Other parameter values are constant across all equations and same as Fig 1d: $p^a=1.11$, $f(q_i)=3.8(q_i)$, l=1, w=5, t=.45, c=15. Sample parameters chosen so that clearing equals 20% across all four cases.

Figure 3: Illustration of 5 km x 5 km microlandscapes and forest fragmentation over time



a) 1973 forest cover

b) 2000 forest cover





Figure 4: Variation in the timing of protected area designations (North and Northeast Thailand)

Source: IUCN Database of Protected Areas and Thai National Parks Division





Graphs show estimated marginal effects of protection from the fixed effects regression in Equation 5 and Table 5, with added interactions allowing the effect to vary by distance to park boundary (interactions added between share protected as NP and WLS and distance, distance² and distance³).

Table 1: Comparison of predicted forest cover outcomes and fragmentation metrics by enforcement type

Outcomes	Forest cover / size (compare protection)	forest patch ed to no	Fragmentation metrics conditional on forest cover (compared to uniform enforcement case)				
	(1)	(2)	(3)	(4)	(5)	(6)	
Enforcement type	All	Interior areas	Forest patch size	Cleared patch size	Forest patch perimeter to area ratio	Density of cleared patches	
Spatially uniform	++	++	0	0	0	0	
Boundary-focused	++	+	-	-	+	+	
Core-focused	++	+++	+	+	-	-	
+ Agglomeration penalty	++	+	-	-	+	+	
+ Labor immobility	+	-	-	-	+	+	

Table summarizes the directional predictions from von Thünen framework in Section 3. The top panel compares three enforcement types: spatially uniform, boundary-focused, and core-focused. The bottom panel considers additional effects from adding an agglomeration penalty or assuming that labor is relatively immobile. Recall that less fragmentation corresponds to more forest cover and larger patch sizes in the interior. Conditional on forest cover, less fragmentation corresponds to larger forest and cleared patch sizes, smaller forest patch perimeter to area ratios and a lower density of cleared patches.

Year	1973	1985	1992	2000
Source	TRFIC*	TRFIC*	TRFIC*	Thai RFD‡
Satellite sensor	Landsat MSS	Landsat MSS	Landsat TM	Landsat TM
type				
Satellite	~60 m	~60 m	~30 m	~30 m
resolution				
Original data	raster	raster	raster	vector
Processing by	georeferenced to	georeferenced to	georeferenced to	processed and
author	match year 2000	match year 2000	match year 2000	georeferenced by
	features; spline	features; spline	features; spline	Thai RFD
	transformation	transformation	transformation	
	(RMS error 189.4) †	(RMS error 196.0) †	(RMS error 274.7) †	

Table 2: Land cover data sources and remote sensing technologies

* Tropical Rain Forest Information Center, Michigan State University. These data were created by a project supported by NASA's Landsat Pathfinder Humid Tropical Forest Project (HTFP) (Samek et al. 2004). The goal was to monitor forest cover in Southeast Asia across time for the purposes of calculating deforestation rates and assessing contributions to global carbon stocks. A detailed description of the process of combining the images into mosaics and interpreting them can be found in Skole et al. 1998.

‡ Thai Royal Forestry Department data; made available online by Marc Souris, IRD.

[†] The 1973-1992 dataset was originally collected for large area analysis. To ensure a higher degree of local spatial accuracy, I geo-referenced each of the layers to match the year 2000 land-use layer. More than 200 control points were selected for each year based mainly on water bodies that can be seen across time in the images. Images were then rectified using a spline transformation. Similar methodology has been used in previous assessments of forest change across time in tropical countries (see e.g. Sanchez-Azofeifa et al. 2001, Viña et al. 2004). Some error does remain after this process, this is summarized by the root mean squared error terms in Table 1. This is classical measurement error and should not bias the regression coefficients.

Variable	Description	Source						
Protection variables								
Share WLS	Share of microlandscape protected as wildlife sanctuary (IUCN I)	IUCN World Database on Protected Areas (2007)†						
Share NP	Share of microlandscape protected as national park (IUCN II)							
Matching variables								
Avg. slope	Average slope (degrees)	National Geospatial						
Avg. elevation	Average elevation (meters)	Intelligence Agency-Digital						
Max. slope	Maximum slope (degrees)	Terrain Elevation Data from						
Max. elevation	Maximum elevation (meters)	USGS Global GIS (1999)						
Distance major city	Distance to nearest major city (pop > 100,000)	ESRI World Cities (2000)						
Dist. 1962 major road	Distance to 1962 major road (km)	Digitized East Asia Road Map,						
Dist. 1962 any road	Distance to 1962 minor road (km)	U.S. Map Service (1964); data from 1962						
Distance rail	Distance to railroad line (km)	Vector Map Level 0 / USGS						
		Global GIS (1997)						
Distance major river	Distance to major river (km) (flow accumulation >	USGS EROS Data Center,						
	5000)	Hydro 1k dataset						
Distance border	Distance to Thai national border (km)	Vector Map Level 0 / USGS						
		Global GIS (1997)						
Ecoregion 2	Share tropical and sub-tropical coniferous forest	WWF Conservation Science						
Ecoregion 3	Share tropical and sub-tropical dry broadleaf forest	Program / USGS Global GIS						
Waterfall	One or more waterfall points in microlandscape	Mapguide Thailand (2009)						

Tuble 51 Data boulees for protected areas and other covariates	Ta	ıb	le	3:	D	ata	sou	rces	for	pro	otect	ed	areas	and	othe	er	covari	iates
--	----	----	----	----	---	-----	-----	------	-----	-----	-------	----	-------	-----	------	----	--------	-------

[†]Years of establishment for protected areas were cross-checked with information from Thailand's Department of National Parks.

Table 4: Summary statistics (matched sample[†])

-

	1973	1985	1992	2000
Protected area (share of microlandscape):				
WLS (IUCN I)	0.052	0.127	0 160	0 196
NP (IIICN II)	0.034	0.094	0.195	0.284
	0.054	0.074	0.175	0.204
Forest cover (share of microlandscape)				
All microlandscapes (N=3510)	0.947	0.824	0.798	0.774
Any protection:				
WLS-interior (N=334)	0.993	0.953	0.960	0.947
WLS-exterior (N=693)	0.945	0.794	0.778	0.785
NP-interior (N=329)	0.982	0.924	0.923	0.924
NP-exterior (N=1255)	0.937	0.813	0.784	0.767
Fragmentation metrics:				
Average forest patch size (sq km)	20.93	16.26	11.46	13.54
Maximum forest patch size (sq km)	23.39	20.19	19.32	18.58
Forest patch perimeter to area ratio (km/km ²)	7.843	15.53	52.72	20.80
Average cleared patch size (sq km)	0.367	1.700	1.343	2.153
Density cleared patches (# patches/ sq km)	0.103	0.205	0.492	0.289
Ν	3510	3510	3510	3510
- ·	2010	0010	0010	2010

Fixed characteristics:	Any I	Protection	No P	rotection	Normdiff*
	Mean	SD	Mean	SD	
Avg. slope (degrees)	7.047	3.573	6.442	3.314	0.124
Avg. elevation (m)	644.3	277.0	595.7	277.1	0.124
Max. slope (degrees)	14.05	6.691	12.89	5.867	0.131
Max. elevation (m)	867.5	342.1	800.7	334.6	0.140
Distance major city (km)	112.7	40.02	115.3	41.04	-0.045
Dist. 1962 major road (km)	19.98	17.98	15.83	14.47	0.180
Dist. 1962 any road (km)	26.46	20.64	19.01	15.03	0.292
Distance rail (km)	87.67	38.18	85.60	41.24	0.037
Distance major river (km)	35.70	20.50	29.17	16.38	0.249
Distance border (km)	61.69	44.04	63.13	45.19	-0.023
Ecoregion 2 (share)	0.018	0.118	0.021	0.134	-0.020
Ecoregion 3 (share)	0.340	0.451	0.447	0.483	-0.161
Waterfall (dummy)	0.037	0.189	0.030	0.172	0.026
Ν	2459		1051		

[†]Matched control microlandscapes (5 km x 5 km) chosen by one-to-one Mahalanobis matching on the covariates listed above. Matching is conducted with replacement and is done separately for the North and Northeast regions. The beginning universe of microlandscapes includes all those which were complete grid cells, which had more than 50% of forest cover in 1973, which had less than 10% cloud cover in any image and less than 20% water in 1973.

*Normdiff is the difference in average covariate values, normalized by the standard deviation (Imbens and Wooldridge 2009).

	Forest cover		Average fore	est patch size	Maximum forest patch size	
	(1)	(2)	(3)	(4)	(5)	(6)
Share PA	0.187*** (0.041)		0.245***		0.210^{***}	
Share WLS	(0.011)	0.220***	(0.011)	0.325***	(0.010)	0.239***
Share NP		(0.074) 0.169*** (0.038)		(0.068) 0.200*** (0.050)		(0.071) 0.193*** (0.037)
Microlandscape fixed effects	Y	Y	Y	Y	Y	Y
Province-year fixed effects	Y	Y	Y	Y	Y	Y
N	14040	14040	14040	14040	14040	14040
Аај к	0.287	0.287	0.402	0.402	0.329	0.329
F-test (WLS=NP)		0.447		0.152		0.478

Table 5: Protected area effects on forest cover and forest patch size (5 km x 5 km microlandscapes)

Table shows coefficients and standard errors from the fixed effects regression model in Equation 5. *** p < .01 ** p < .05 * p < 10. Standard errors (in parentheses) are robust and clustered at the district level.

The dependent variables are the log of each outcome. Protection is measured as the share of each microlandscape protected. Fixed effects are included for each microlandscape and for province-year, to capture regional deforestation trends.

	Forest cover		Average for	est patch size	Maximum fo	prest patch size
	(1)	(2)	(3)	(4)	(5)	(6)
Share PA	0.220***		0.314***		0.319***	
Share WLS	(0.000)	0.251**	(0.105)	0.575***	(0.000)	0.350***
Share NP		(0.100) 0.202*** (0.075)		(0.116) 0.168 (0.129)		(0.115) 0.301*** (0.097)
Microlandscape fixed effects	Y	Y	Y	Y	Y	Y
Province-year fixed effects	Y	Y	Y	Y	Y	Y
N	4384	4384	4384	4384	4384	4384
Adj R ²	0.387	0.387	0.521	0.522	0.413	0.413
F-test (WLS=NP)		0.633		0.027		0.654

Table 6: Protected area effects on forest cover and forest patch size (10 km x 10 km microlandscapes)

Table shows coefficients and standard errors from the fixed effects regression model in Equation 5. *** p < .01 ** p < .05 * p < 10. Standard errors (in parentheses) are robust and clustered at the district level.

The dependent variables are the log of each outcome. Protection is measured as the share of each microlandscape protected. Fixed effects are included for each microlandscape and for province-year, to capture regional deforestation trends.

	Average forest patch size	Max forest patch size	Average cleared patch size	Forest patch perimeter to area ratio	Cleared patch density
	(1)	(2)	(3)	(4)	(5)
Share WLS	0.052 (0.032)	0.004 (0.013)	0.082*** (0.028)	-0.167** (0.075)	-0.011 (0.016)
Share NP	-0.023	0.001	-0.009	0.076	0.040**
Forest cover	3.080*** (0.117)	2.657*** (0.080)	-2.499*** (0.072)	-1.565*** (0.353)	0.066 (0.063)
Microlandscape fixed effects	Y	Y	Y	Y	Y
Province-year fixed effects	Y	Y	Y	Y	Y
N	14040	14040	14040	14040	14040
Adj R ²	0.669	0.868	0.659	0.324	0.339
F-test (WLS=NP)	0.084	0.840	0.005	0.020	0.002

Table 7: Protected area effects on fragmentation metrics, conditional on forest cover (5 km x 5 km microlandscapes)

Table shows coefficients and standard errors from the fixed effects regression model in Equation 5, adding forest cover as a control.

*** p < .01 ** p < .05 * p < 10. Standard errors (in parentheses) are robust and clustered at the district level.

The dependent variables are the log of each outcome. Protection is measured as the share of each microlandscape protected. Fixed effects are included for each microlandscape and for province-year, to capture regional deforestation trends.

	Average	Max forest	Average	Forest patch	Cleared patch
	forest patch	patch size	cleared patch	perimeter to	density
	size		size	area ratio	
	(1)	(2)	(3)	(4)	(5)
Share WLS	0.257**	0.033	0.214***	-0.236	-0.017
	(0.109)	(0.054)	(0.071)	(0.162)	(0.022)
Share NP	-0.083	0.051	-0.011	0.140	0.061**
	(0.116)	(0.040)	(0.077)	(0.194)	(0.025)
Forest cover	3.156***	3.148***	-2.993***	-0.053	0.141**
	(0.212)	(0.187)	(0.165)	(0.365)	(0.065)
Microlandscape	Y	Y	Y	Y	Y
fixed effects					
Province-year	Y	Y	Y	Y	Y
fixed effects					
Ν	4384	4384	4384	4384	4384
$Adj R^2$	0.615	0.775	0.570	0.405	0.419
·					
F-test	0.042	0.730	0.009	0.137	0.003
(WLS=NP)					

Table 8: Protected area effects on fragmentation metrics, conditional on forest cover (10 km x 10 km microlandscapes)

Table shows coefficients and standard errors from the fixed effects regression model in Equation 5, adding forest cover as a control.

*** p < .01 ** p < .05 * p < 10. Standard errors (in parentheses) are robust and clustered at the district level.

The dependent variables are the log of each outcome. Protection is measured as the share of each microlandscape protected. Fixed effects are included for each microlandscape and for province-year, to capture regional deforestation trends.